Author’s response to reviews

Title: Daytime use of general practice as a predictor for contacts to Out-of-Hours services in primary healthcare for patients with chronic disease: a cohort study

Authors:

Lone LF Flarup (l.flarup@alm.au.dk)
Grete GM Moth (g.moth@alm.au.dk)
Morten MBC Christensen (mbc@alm.au.dk)
Mogens MV Vestergaard (mogens.vestergaard@alm.au.dk)
Frede FO Olesen (fo@alm.au.dk)
Peter PV Vedsted (p.vedsted@alm.au.dk)

Version: 3
Date: 17 June 2014

Author’s response to reviews: see over
Dear editor,

Aarhus d. 17 June 2014

Concerning the manuscript MS: 4784551651316707, ‘Daytime use of general practice as a predictor for contacts to Out-of-Hours services in primary healthcare for patients with chronic disease: a cohort study’

We have revised the manuscript carefully according to the editorial comments.

Analyses are now adjusted for clustering by practice using robust variance estimates. This adjustment did, however, not affect the estimates.

Yours sincerely,

Lone Flarup